Author’s response to reviews

Title: Brain biomarkers and pre-injury cognition are associated with long-term cognitive outcome in children with traumatic brain injury

Authors:

Amy Wilkinson (amy.wilkinson@sickkids.ca)
Maureen Dennis (maureen.dennis@sickkids.ca)
Nevena Simic (nsimic@hhsc.ca)
Margot Taylor (margot.taylor@sickkids.ca)
Benjamin Morgan (ben.morgan@sickkids.ca)
Helena Frndova (Helena.frndova@sickkids.ca)
Karen Choong (choongk@mcmaster.ca)
Craig Campbell (craig.campbell@lhsc.on.ca)
Douglas Fraser (douglas.fraser@lhsc.on.ca)
Vicki Anderson (vicki.anderson@rch.org.au)
Anne-Marie Guerguerian (anne-marie-guerguerian@sickkids.ca)
Russell Schachar (Russell.schachar@sickkids.ca)
Jamie Hutchison (jamie.hutchison@sickkids.ca)

Version: 2 Date: 09 Jul 2017

Author’s response to reviews:

We thank the reviewer and editor for his/her helpful questions, comments and suggestions. We have revised our Author's Contributions section to include more details. It reads as follows:

JH, RS, MD, AMG and VA conceptualized, designed and led the research project and were responsible for acquiring funding. AMG, JH, KC, CC and DF were responsible for and supervised research coordinators who collected data at the 3 study sites. MD, AW and NS conceptualized and designed the follow-up portion of the study. AW collected all follow-up data, analyzed and interpreted data and drafted the manuscript. HF created and managed the database for the study. BM and MT performed the PLS analyses and guided the interpretation of the
results. All authors critically reviewed and revised and then approved the final manuscript as submitted. All authors are accountable for all aspects of the work and gave final approval for publication.